Author's response to reviews

Title: Leg orientation as a clinical sign for pusher syndrome

Authors:

Leif Johannsen (l.johannsen@bham.ac.uk)
Doris Broetz (doris.broetz@med.uni-tuebingen.de)
Hans-Otto Karnath (karnath@uni-tuebingen.de)

Version: 6 Date: 21 August 2006

Author's response to reviews:

The following format changes as requested by the BioMed Central Editorial Production Team have been worked into the manuscript:
1. the keywords section has been removed from the manuscript
2. the running title has been removed from the title page
3. figure and table placement text has been removed
4. figure section has been named ‘Figure legends’
5. image files have been renamed
6. figures have been closely cropped
6. figure orientation has been changed to portrait